<%namespace name="tools" file="tools.tex"/>
\textbf{Results}\newline
The comparison between the observed and numerically predicted beach profiles with the BOI parameter settings is shown in \autoref{fig:T01_BOI},  \autoref{fig:T02_BOI}, \autoref{fig:T03_BOI} and \autoref{fig:T08_BOI}. The observed profiles are represented by a dashed line and the computed profiles through a solid line. The line color indicates the moment in time. \autoref{tab:Deltaflume2006_T01_indicator} to \autoref{tab:Deltaflume2006_T08_indicator} show the results for the different indicators. In \autoref{fig:T01_BOI_errorTime}, \autoref{fig:T02_BOI_errorTime}, \autoref{fig:T03_BOI_errorTime} and \autoref{fig:T08_BOI_errorTime}, the relative error is plotted as a function of time.  Besides a comparison of the observed and predicted beach profiles, also a detailed hydrodynamic and morphodynamic analysis is performed, which is presented in Appendix \ref{sec:Detailed_analysis}.

${tools.image('Deltaflume2006/T01/', 'T01_BOI_profiles.eps', 'Comparisons of profiles from experiment T01 for different moments in time. Observed profiles are shown with a dashed line and the XBeach profiles with a solid line. The storm surge level is shown with a black dashed line.',label='T01_BOI',width=150)}
${tools.table('Deltaflume2006/T01', 'indicator.tex', 'Indicators for test T01',small=True)} 
${tools.image('Deltaflume2006/T01/', 'T01_BOI_errorTime.eps', 'Temporal development of the relative error for the three indicators.',width=100,label='T01_BOI_errorTime')}

${tools.image('Deltaflume2006/T02/', 'T02_BOI_profiles.eps', 'Comparisons of profiles from experiment T02 for different moments in time. Observed profiles are shown with a dashed line and the XBeach profiles with a solid line. The storm surge level is shown with a black dashed line.',label='T02_BOI',width=150)}

${tools.table('Deltaflume2006/T02', 'indicator.tex', 'Indicators for test T02',small=True)} 
${tools.image('Deltaflume2006/T02/', 'T02_BOI_errorTime.eps', 'Temporal development of the relative error for the three indicators.',label='T02_BOI_errorTime')}


${tools.image('Deltaflume2006/T03/', 'T03_BOI_profiles.eps', 'Comparisons of profiles from experiment T03 for different moments in time. Observed profiles are shown with a dashed line and the XBeach profiles with a solid line. The storm surge level is shown with a black dashed line.',label='T03_BOI',width=150)}
${tools.table('Deltaflume2006/T03', 'indicator.tex', 'Indicators for test T03',small=True)} 
${tools.image('Deltaflume2006/T03/', 'T03_BOI_errorTime.eps', 'Temporal development of the relative error for the three indicators.',label='T03_BOI_errorTime')}


${tools.image('Deltaflume2006/T08/', 'T08_BOI_profiles.eps', 'Comparisons of profiles from experiment T08 for different moments in time. Observed profiles are shown with a dashed line and the XBeach profiles with a solid line. The storm surge level is shown with a black dashed line.',label='T08_BOI',width=150)}
${tools.table('Deltaflume2006/T08', 'indicator.tex', 'Indicators for test T08',small=True)} 
${tools.image('Deltaflume2006/T08/', 'T08_BOI_errorTime.eps', 'Temporal development of the relative error for the three indicators.',label='T08_BOI_errorTime')}

\textbf{Overview}\newline
An overview of the statistical scores is shown in \autoref{fig:DG2006_overview} and \autoref{tab:DG2006_overview}.

${tools.image('Deltaflume2006/', 'overview.eps', 'Overview of statistical scores for the Deltagoot 2006 experiments.',label='DG2006_overview')}

${tools.table('Deltaflume2006/', 'skill.tex', 'Statistical scores for Deltagoot 2006 experiments', 'DG2006_overview')} 